Author's response to reviews

Title: Willis-Ekbom Disease Is not Associated With Poor Cardiovascular Health in Adults.

Authors:

  Brynn K Dredla (Dredla.Brynn@mayo.edu)
  Oscar H Del Brutto (oscardelbrutto@hotmail.com)
  Augustine S Lee (Lee.Augustine@mayo.edu)
  Pablo R Castillo (castillo.pablo@mayo.edu)

Version: 3 Date: 28 October 2015

Author's response to reviews: see over
October 28, 2015

Dr Bjorn R. Olsen, Editor-in-Chief  
Journal of Negative Results in Biomedicine  
236 Gray’s Inn Rd  
London WC1X 8HB  
United Kingdom

Dear Doctor Olsen:

RE: Willis-Ekbom Disease is Not Associated With Poor Cardiovascular Health in Adults by Brynn k. Dredla, MD, Oscar H. Del Brutto, MD, Augustine S. Lee, MD, and Pablo R. Castillo, MD

Attached is our revised manuscript, which addresses the suggestions from the reviewer.

We deleted a sentence from the Methods section, as suggested by the reviewer.

We also replaced 2 words from the title, using “Adults” (instead of "Elderly Persons") for accuracy, because the age in our study was older than 40 years.

Regards,

Pablo R. Castillo, MD